Author's response to reviews

Title: No connection between the level of exposition to statins in the population and the incidence/mortality of acute myocardial infarction: An ecological study based on Sweden’s municipalities.

Authors:

Staffan Nilsson (staffan.nilsson@lio.se)
Sigvard Mölstad (sigvard.molstad@lj.se)
Catarina Karlberg (catarina.karlberg@vgregion.se)
Jan-Erik Karlsson (jan-erik.karlsson@lj.se)
Lars-Göran Persson (lg.persson@lj.se)

Version: 4 Date: 23 May 2011

Author's response to reviews: see over
Dear Editor,
Revisions have now been made according to your list.

Major revisions
References:
I have checked the link to the reference Sveriges Kommuner och Landsting [http://www.skl.se/web/Aldre_indelning_1999-2004.aspx] several times. The link does work. However it is provided only in Swedish.

Minor comments
All highlightings have been removed from the manuscript.

Abbreviations are now in a single paragraph and the title of the paragraph has been changed accordingly.

Figures: The figure numbers have been removed from the image files.

‘Methods’ is now placed directly after ‘Background’. As a consequence the numbering of the reference list has been changed accordingly.

The 'Reference' title is now placed directly at the start of the reference list.

Final check of typographical errors has been done.

Affiliations on the front page have been updated.

Best wishes

Staffan Nilsson